Author’s response to reviews

Title: Growing up HIV-positive in Uganda: "psychological immunodeficiency?" A qualitative study

Authors:

Birthe Loa Knizek (birthe.l.knizek@ntnu.no)

James Mugisha (jmmugi77@hotmail.com)

Joseph Osafo (josaforo@gmail.com)

Eugene Kinyanda (Eugene.Kinyanda@mrcuganda.org)

Version: 1 Date: 03 Feb 2017

Author’s response to reviews:

Dear editor,

Thank you for giving us the possibility to revise. We have done so according to your advice. We have now

1) included in the methods section information according to the CoreQ gudidelines

2) included the COREQ guidlines scheme

3) explained why the data are not available

4) included all necessary declarations at the end

We hope that the manuscript now has a shape, which is worthy to be sent out to reviewers.

Sincerely,

Birthe Loa Knizek